

Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 16438

Low-Wage Jobs, Foreign-Born Workers, and Firm Performance

Catalina Amuedo-Dorantes Esther Arenas-Arroyo Parag Mahajan Bernhard Schmidpeter

SEPTEMBER 2023



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 16438

Low-Wage Jobs, Foreign-Born Workers, and Firm Performance

Catalina Amuedo-Dorantes University of California Merced and IZA

Esther Arenas-Arroyo Vienna University of Economics and Business and IZA **Parag Mahajan** University of Delaware

Bernhard Schmidpeter Johannes Kepler University, RWI and IZA

SEPTEMBER 2023

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Low-Wage Jobs, Foreign-Born Workers, and Firm Performance^{*}

We examine how migrant workers impact firm performance using administrative data from the United States. Exploiting an unexpected change in firms' likelihood of securing lowwage workers through the H-2B visa program, we find limited crowd-out of other forms of employment and no impact on average pay at the firm. Yet, access to H-2B workers raises firms' annual revenues and survival likelihood. Our results are consistent with the notion that guest worker programs can help address labor shortages without inflicting large losses on incumbent workers.

JEL Classification:	J23, F22, J61
Keywords:	guest workers, migrants, employment, firm dynamics, H-2B visa

Corresponding author:

Esther Arenas-Arroyo Department of Economics Vienna University of Economics and Business (WU) Welthandelsplatz 1 1020 Vienna Austria E-mail: esther.arenas.arroyo@wu.ac.at

^{*} We are extremely grateful to Joe Ballegeer of the Philadelphia Federal Statistical Research Data Center for help in disclosing results. We also thank Ethan Lewis, Nicolas Morales, Bryan Stuart, and participants at the Society of Labor Economics Annual Meetings and Annual Washington Area Labor Economics Symposium for helpful comments. Any views expressed are those of the authors and not those of the U.S. Census Bureau. The Census Bureau has reviewed this data product to ensure appropriate access, use, and disclosure avoidance protection of the confidential source data used to produce this product. This research was performed at a Federal Statistical Research Data Center under FSRDC Project Number 2105. (CBDRB-FY23-0279)

1 Introduction

Firms increasingly report difficulties finding workers, particularly in sectors that employ low-wage workers.¹ In 2018, roughly 25% of EU firms in the construction and service sectors reported that labor shortages limit their production, compared to less than 5% in 2010 (European Commission, 2022). In the US, job opening rates in the construction and the food and accommodation sectors more than doubled over the same time period. In light of these developments, calls for increasing (temporary) immigration have become commonplace.² Yet, we still have a limited understanding of how guest workers impact firm hiring performance, particularly when labor markets are tight.

The US H-2B visa program offers a unique opportunity to address this question. It is the primary channel through which US firms secure guest workers for non-agricultural jobs that do not require a college degree. It therefore tends to match workers who make less than \$15 per hour with large, locally important firms.³ Because it risks displacing especially vulnerable native workers, the H-2B visa program has been the subject of a contentious policy discussion that mirrors the larger debate surrounding US immigration. On the one hand, users of the program argue that having access to foreign-born labor is vital to their survival, pointing out that they face a shortage of US workers willing to perform the jobs they offer. On the other hand, opponents of the H-2B visa program argue that firms primarily use the H-2B program to save on labor costs.⁴

In this paper, we use a unique suite of administrative data to study how the H-2B visa program impacts firm performance. We compile a data set that links administrative data on 1) firm initial requests for H-2B workers from the Department of Labor (DoL), 2) firm-level counts of eventual H-2B visa approvals from the United States Customs and Immigration Services (USCIS), and 3) administrative data on firm-level outcomes from the US Census Bureau. Our final data set includes the near-universe of H-2B applicant firms. It allows us to observe information on their quarterly payroll, quarterly employment, and annual revenues over the period spanning from 2015 through 2019.

¹See, e.g., Kosakow and Waddell (2022) for a recent survey of employers that makes this point.

²See, e.g., Gordon H. Hanson and Matthew J. Slaughter, "America Needs More Immigration to Defeat Inflation" in *Foreign Affairs*; Vanessa Yurkevich, "America needs immigrants to solve its labor shortage" from CNN; Justin Gest, "How Immigrants Tame Inflation" from the *Wall Street Journal*.

³That these workers are therefore at the lower end of the hourly wage distribution is the sense in which we use the term "low-wage" in this paper. See Section 4.1 for more on H-2B employers and the wages they offer.

⁴See, for example for congressional testimony by a large, Michigan hotel owner in favor of the H-2B program: https://www.visalawyerblog.com/Musser080416H2B%20testimony.pdf. Several popular press accounts also highlight the importance of immigrant workers to firms in lower-wage sectors. See, e.g., Miriam Jordan and Santiago Pérez, "Small Businesses Lament There Are Too Few Mexicans in US, Not Too Many" in the *Wall Street Journal* and Jeff Barker, "Visas for crab processors is a 'one-year remedy" in the *Washington Post*. For a summary of arguments made by H-2B detractors, see Daniel Costa, "Claims of labor shortages in H-2B industries don't hold up to scrutiny".

The primary challenge in assessing how access to H-2B workers impacts firms is that firms participating in the H-2B program are likely selected along various unobserved dimensions. To identify the effect of access to H-2B workers on firm performance, we exploit the unprecedented spike in Temporary Labor Certification (TLC) applications for H-2B workers that occurred on January 1, 2018. This spike forced the DoL to change its processing procedures ex-post, generating quasi-random variation in firms' access to H-2B workers workers during the second half of fiscal year 2018 (calendar Q2 and Q3 of 2018). Specifically, firms that applied to the DoL before 7:00 am EST on January 1, 2018 (early-applicants) were much more likely to have their applications processed in time to participate in on-time H-2B hiring than firms that applied later on January 1, 2018 (late-applicants). On average, early-applicants gained access to approximately eight additional H-2B workers relative to late-applicants. Yet, because this cutoff was not predictable ex-ante, early-applicants and late-applicants were on parallel trends prior to 2018, prompting us to use a difference-in-differences approach to identify how access to low-wage, foreign-born workers impacted firm outcomes in 2018 and 2019.^{5,6}

We start by exploring the canonical question of whether hiring foreign-born workers through the H-2B program crowds out other forms of employment at the firm level. We find that employment increased significantly at early-applicant firms during the H-2B hiring period of calendar Q2 and Q3 in 2018, by roughly 0.63 employees per H-2B approval and 0.81 employees per actual H-2B hire. In the fourth quarter, when most H-2B workers were no longer employed, estimates revert back to zero. Given that these estimates likely include the H-2B workers themselves, we find limited scope for crowd-out of other workers (0.81 - 1 = -0.19 other workers per H-2B hire).⁷ These results are consistent with the notion that firms are mostly hiring H-2B workers to tackle seasonal labor shortages.

Second, we assess whether firms are using the H-2B program to cut down on labor costs, as argued by some of the program's opponents. Using information on quarterly payroll, we fail to find any evidence that H-2B workers reduce average (per worker) or total payroll costs among hiring firms. Instead, we find payroll increases that imply small, positive spillover effects on the non-H-2B wage bill. We do not find any effects on longer-term payroll increases in 2019.

⁵The 7:00 am EST "cutoff" may bring to mind a regression discontinuity design, i.e., Pinoti (2017). Unfortunately, we observe very few firms around the threshold, which leads to small sample problems. Instead, we adopt a difference-in-differences approach that allows us to use the entire data set, while providing evidence that the parallel trends and the no anticipation assumptions hold in our setting.

⁶There are a small number of firms who sent in applications both before and after 7:00 am EST (for different sets of workers). Therefore, as we further explain in Section 3, our treatment is technically continuous. However, given that this group is small, we discuss our treatment in binary terms here to facilitate exposition.

⁷As H-2B workers are subject to federal income taxes, they are counted in the data as employees. Since we do not observe firm workforce composition by nativity in our data, we cannot explore whether the crowd-out affects foreign-born or natives.

Third, we assess the impact of access to H-2B workers on key measures of firm performance. We find that access to H-2B workers significantly increases firm revenues, with early-applicants netting an additional \$9.5 thousand in revenues relative to early-applicants in 2018. Furthermore, we find that firms with access to H-2B workers exhibit a higher likelihood of staying active as an employer, as measured by an indicator for having a positive annual payroll. Specifically, early-applicant firms had a 1.6 percentage point higher likelihood of being active in 2018 and a 3 percentage point higher likelihood in 2019. These large extensive margin effects indicate that, among H-2B participant firms, profitability and viability may hinge on the ability to hire foreign-born workers.

Heterogeneity analyses reveal that firms with higher initial revenues per worker and in stricter immigration enforcement environments benefit more from access to H-2B workers, indicating that higher productivity firms in more constrained hiring environments may be most reliant on the program. Lastly, we do not find strong evidence of H-2B participant firms in markets with more early-applicants faring differently from H-2B participant firms in markets with fewer early-applicants, indicating limited potential for contamination in our estimates from cross-firm spillovers.

Our study contributes to the existing literature on the impact of immigration on labor markets. Most studies evaluating the impact of "low-skill" immigration focus on worker-level outcomes or market-level outcomes derived from worker-level data (e.g., Borjas, 2003; Dustmann et al., 2017; Clemens et al., 2018; Abramitzky et al., 2022; East et al., 2023). In contrast, most of the research using firm-level data focuses on the hiring of "high-skilled" immigrant workers, mainly through the H-1B program (e.g., Kerr and Lincoln, 2010; Peri, 2012; Pekkala Kerr et al., 2015; Peri et al., 2015; Doran et al., 2022; Brinatti et al., 2023) or on market-level immigration shocks (e.g., Dustmann and Glitz, 2015; Mitaritonna et al., 2017; Ayromloo et al., 2020; Orefice and Peri, 2020; Beerli et al., 2021; Brinatti and Morales, 2021; Mahajan, 2022). A singular exception is Clemens and Lewis (2022), who examine the effect of the 2021 H-2B visa lottery using a survey of participant firms.⁸ They provide evidence that winning the H-2B lottery leads to a significant expansion in production, revenues, investment, and employment. They also rule out substantive native displacement.

Our work complements and extends these findings in important ways. First, we confirm some of the primary findings in Clemens and Lewis (2022)—particularly the increase in firm revenues and relative lack of employment crowd-out—using the full set of potential H-2B users, a different year, and a different methodological approach. This is particularly important given that the 2018 change to DoL processing that we exploit was unanticipated by firms, whereas the 2021 lottery studied in Clemens and Lewis (2022) was

⁸The DoL began holding lotteries that determined which TLC applications were processed first starting in 2020. These lotteries were implemented due to the application spike we study in 2018 and a subsequent application spike that took place in 2019.

fully anticipated. The similar results, then, further buttress the case that H-2B workers are essential to the operation of H-2B users.

Second, the use of comprehensive, administrative firm-level data allows us to add important dimensions to our understanding of the impact of foreign-born workers on firms. For example, our results on firm survival speak directly to the essentiality of H-2B workers to a set of large, locally important firms. Our results on payroll provide direct evidence that guest workers who earn low wages do not necessarily price incumbent workers out of jobs. Instead, we find evidence of a temporary pay spillover to non-H-2B workers. In this regard, our results are in contrast to recent findings on the impact of H-1B visa workers on firms in Doran et al. (2022). They find that these college-educated, migrant workers significantly crowd out other workers and decrease average earnings at the firm. The differences across these studies underscore the distinct nature and impact of foreign-born employment in different parts of the wage distribution and foreground the notion that recent labor shortages may be particularly acute in lower-paying sectors.

The rest of this paper proceeds as follows. Section 2 describes relevant details of the H-2B visa program. Section 3 describes our identification strategy. Section 4 describes the H-2B data, the administrative firm-level data, and the matching of these two data sources. Section 5 provides evidence consistent with our identification strategy, while Section 6 presents our main findings. Section 7 examines heterogeneous responses and presents a test for spillover effects across firms. Section 8 concludes the paper.

2 Institutional Setting and Labor Market Context

2.1 Overview of the H-2B Visa Program

The H-2B visa program has its roots in the World War II era when the War Food Administration recruited guest workers from various Central and South American countries through the Bracero program.⁹ As time passed, the Bracero program underwent several changes until the 1986 Immigration Reform and Control Act divided it into two separate visa programs: H-2A for agricultural workers and H-2B for non-agricultural workers.

The H-2B visa program serves as a means for US employers to hire foreign-born individuals to fulfill temporary and full-time non-agricultural positions within a specified area(s) of intended employment.¹⁰ Jobs offered through this program typically have a

⁹For a brief review of the program, please visit: https://guides.loc.gov/latinx-civil-rights/bracero-program.

¹⁰H-2B visas belong to the category of nonimmigrant visas granted to foreign nationals who seek temporary entry into the United States. Alongside H-2B visas, there are several other nonimmigrant visas permitting foreign nationals to work in the US for a specific time and purposes. These include H-1B visas, designed for college-educated workers in specialty occupations, H-2A visas for agricultural laborers, B-1 visas catering to business travelers, B-2 visas for tourists, and J visas for exchange visitors like certain teachers

duration of up to nine months and require a minimum of 35 hours of work per week.¹¹ The H-2B program admits a total of 66,000 workers annually, and this number is equally divided between the two halves of the fiscal year.¹²

2.2 Employing an H-2B Visa Worker

Employers seeking to hire foreign-born workers under the H-2B visa program must navigate a multi-step process involving three U.S. Departments: the Department of Labor (DoL), U.S. Citizenship and Immigration Services (USCIS), and the Department of State (DoS). Therefore, many H-2B program users contract out the application process to immigration attorneys. For example, our calculations indicate that 92% of initial H-2B applications to the DoL were submitted by attorneys in 2018. Hired attorneys file applications electronically or send the required documents by mail.





Source: Authors' analysis of DOL, DHS, and State regulations and guidance

Initially, employers must obtain a Temporary Labor Certification (TLC) from the DoL, which requires registration 120 to 150 days before the intended job start date.¹³ To secure the certification, they need to demonstrate the lack of native workers willing, qualified, and able to perform the job, as well as show that hiring an H-2B worker will not adversely affect the wages and conditions of local employees. Before submitting their applications, companies need to determine the prevailing wage rate for the position. This

and students. Of these, only the J visa has the potential to be a substitute for an H-2B visa, based on education requirements and occupational constraints. However, the J visa has primarily been used by firms in the hospitality and leisure sector, whereas H-2B users have increasingly become concentrated in Landscaping.

¹¹While nine months is the maximum, most firms that demand H-2B workers in the second half of a fiscal year only do so during what we observe as calendar Q2 and Q3 of a given year. For example, 71% of applications for H-2B workers to the DoL in 2018 indicated an end date before December 12, 2018, when Q4 employment is measured in our Census firm data.

¹²A fiscal year in the US starts October, 1 and ends September, 30 in the following year. Consequently, the first half of a fiscal year is from October, 1 until March, 30. The second half of a fiscal year starts April 1 and ends September, 30.

¹³TLC applications for the same occupation and worksite are consolidated into a single form.

information is obtained from the National Prevailing Wage Center and involves comparing the pay of non-H-2B workers in the same occupation and geographic area.¹⁴ The H-2B worker's pay must be higher than both the prevailing wage rate and the applicable Federal, State, or local minimum wage.¹⁵ Part of the TLC requirements also includes extensive advertising of the position, contacting former workers and union representatives, and utilizing other suggested channels by the certifying officer. This has to be done 75 to 90 days before the job start date, and the firm is required to accept all qualified US applicants up to 21 days before the job start date.

Due to existing deadlines, the DoL experiences a spike in applications at the beginning of each year for seasonal workers in the spring and summer—the second half of a given fiscal year—when most temporary workers are needed. The earliest a firm can apply for a 2nd Half Fiscal Year (2HFY) worker is January 1, and the earliest this worker can start employment is April 1. Our identification strategy is based on an unusual spike in TLC applications that occurred on January 1, 2018 (see Section 2.4 for details).

Following DoL's approval of their TLC applications, firms can submit official petitions for H-2B workers to the USCIS in an i129 form. The USCIS determines whether there is still room under the statutory cap for the workers requested and then charges a base of \$460 for processing an application. Most firms select a premium service, which costs an additional \$1,500, but guarantees faster processing, i.e., within 15 days.¹⁶ USCIS usually processes applications sequentially, in the order they were received. However, if USCIS estimates that the cap will be exceeded based on the number of petitions received within five business days after the application start date, it conducts a lottery to allocate the H-2B visas randomly.¹⁷

Following USCIS approval, firms can start the hiring process. Each prospective employee has to apply for an H-2B visa at their corresponding US embassies, and the DoS charges employers an additional \$190 per worker for visa processing. The DoS screens the applications and can still refuse to issue some visas. To our knowledge, the only public information on the rate of conversion from a USCIS approval to a DoS approval comes from a DHS report to Congress in 2016.¹⁸ Table B1 shows Department of State H-2B capsubject visa issuances and USCIS H-2B, cap-subject visa approvals for the years available from this report, along with data on DoS denials in years when available.¹⁹ Issuances by

¹⁴Information on the determinants of the prevailing wage rate is given here https://www.dol.gov/ agencies/eta/foreign-labor/wages.

¹⁵See, for example, the Fact Sheet on H-2B wage requirements, available under https://www.dol.gov/ agencies/whd/fact-sheets/78c-h2b-wage-requirements.

¹⁶See the USCIS fee schedule https://www.uscis.gov/sites/default/files/document/forms/ g-1055.pdf. The premium service does not, however, modify the firm's position in the TLC submission queue or enable late-submitting firms priority over firms that submitted their applications earlier.

¹⁷USCIS analyzes full firm petitions instead of worker solicitations. This means that firms' petitions are either fully approved or denied. Therefore, firms have no incentive to apply for more workers than needed.
¹⁸Link here: H-2B Usage and Recommendations.

¹⁹DoS visa denial data available here: DoS Non-Immigrant Visa Statistics.

the DoS are generally well below USCIS approvals, with conversion rates ranging form 57.0% to 88.1%. In practice, this means that a firm hires fewer H-2B workers than it receives approvals for from the USCIS, mostly because of DoS denials. We use 78%—the average conversion rate across years—as our benchmark for the rate at which USCIS H-2B approvals convert into DoS H-2B issuances ("hires").

2.3 The Post-Recession, Pre-Pandemic Labor Market

Demand for H-2B workers has steadily increased since the recovery from the Great Recession. Figure 2 shows the initial number of applications for H-2B workers (roughly, demand) and the final number of visas issued (supply) from 2010 through 2019.²⁰

Two features emerge from Figure 2. First, the demand for H2B-workers always exceeded the number of visas available over the period under consideration. Second, the figure shows a clear increase in the demand for H-2B workers over time. Excess demand mirrors the increase in total demand, given statutory constraints. For example, in 2018, around 80,000 visas were issued, but almost 150,000 applications were received, implying that almost 50% of the initial demand was not met.





Sources: DoL Office of Foreign Labor Certification Performance Data (total LCA applications) and Department of State (visas issued).

The increasing demand for H-2B workers may have been driven by firms' inability to address their labor needs through other channels. In the left panel of Figure 3, we plot the unemployment rate for workers with and without a college degree from 2000 to 2019. The years 2015 through 2019—the period for which we have data available—is marked

²⁰This brackets the period after the Great Recession and before the COVID-19 pandemic. We do not consider the COVID years due to their extraordinary nature. After a drop in demand for H-2B workers during the pandemic, demand for H-2B workers has picked up again in the post-pandemic period. In 2022, requests for H-2B workers were substantially higher than they were even in 2019.

by two vertical lines in each figure. Non-college workers have generally faced higher unemployment rates and have been harder-hit by recessions (shown by the gray-shadowed areas). Since the end of the Great Recession, however, the non-college unemployment rate has witnessed a remarkable decline, falling from 12% at the beginning of 2010 to roughly 4% at the beginning of 2019. In comparison, the unemployment rate of workers with a college degree fell from 4% to 3% over the same period.

The drop in the unemployment rate of non-college workers was accompanied by a strong increase in job openings in industries generally associated with low- and medium-wage jobs, such as food and accommodation and construction. Using data from the Bureau of Labor Statistics Job Openings and Labor Turnover Survey (JOLTS), the right panel of Figure 3 plots job opening rates in low- and medium-wage industries over time. Job openings grew steadily from the end of the Great Recession until the end of our sample period. Kosakow and Waddell (2022) show that firm expectations of wage growth among low-wage workers were higher than those of high-wage workers, even in the prepandemic labor market. In sum, we believe that the increased demand for H-2B workers during our study period is symbolic of the increase in demand for workers in low-wage industries.





Notes: Thick dashed lines connect values in April of each year. Thinner lines represent non-seasonally-adjusted, monthly values. Gray bars represent National Bureau of Economic Research (NBER) recessions. Vertical black lines bracket this paper's study period: January 2015 through December 2019.

Sources: Authors calculations from BLS Current Population Survey (CPS) microdata, via IPUMS-CPS (Ruggles et al., 2022) (left) and Bureau of Labor Statistics (BLS) Job Openings and Labor Turnover Survey (JOLTS) (right) and .

2.4 The 2018 TLC Application Spike

Prior to 2018, the DoL processed applications based on the day they were filed, irrespective of the time of day. This procedure gave firms submitting their TLC applications on the same day a roughly equal chance to proceed to the next stage—i129 submission to the USCIS—on time. Further, while demand for H-2B workers had been steadily rising prior to 2018, the H-2B FY 2nd Half cap was not reached until mid-to late March in 2015, 2016, and 2017. Hence, firms applying on January 1st would have almost certainly received certification in time to submit an i129 petition to the USCIS.

However, on January 1, 2018, the DoL received around 4,500 TLC applications for more than 81,000 positions with an April 1 start date, exceeding the semi-annual visa allotment by almost 250%.²¹ This was the first time that enough TLC applications to fill the H-2B cap were received on the first possible application date. For comparison, the DoL received applications for 26,673 positions on January 1, 2017 and less than 10,000 positions on January 1, 2016. Following the surge in applications, the DoL announced modifications to the adjudication process on January 17, 2018. Certified applications would be released on February 20, 2018 based on the exact day *and* time—down to the millisecond—that they were received.²² Anecdotal evidence,²³ along with direct evidence we provide in Section 6, suggests that firms did not anticipate this change to DoL procedures.

After the January 17 announcement on changes to its procedures, the DoL started to process and release approved TLC applications, i.e., on February 20, 2018. Within five business days, the USCIS had already received 2,700 i129 applications for a total of 47,000 workers, well-exceeding the 33,000 cap. USCIS conducted its first ever lottery for i129 petitions for H-2B visas on February 28, 2018. In addition, USCIS announced that further petitions would be denied.²⁴

Given that approved TLC applications are required for i129 petitions, the change to the DoL processing generated variation in firms' ability to send their i129 petitions to USCIS in time to *participate* in the lottery. As a comparison, in FY2017, 98.5% of TLC applications received on January 1 were processed in time for firms to send their certified i129 applications before the cap was reached. In FY2018, that figure dropped to 71.4%. Applications filed one hour later on January 1, 2018 were processed 0.6 days later and resulted in a 6 percentage point drop in the probability of receiving a TLC approval in time to participate in the lottery.²⁵

Figure 4 shows that the decline in on-time processing over the course of applications received on January 1 was not linear. Applications submitted prior to 7:00am were substantially more likely to be processed before February 27 relative to those received later in the day. Specifically, 96% of TLC applications received before 7:00am were

²¹See the January 23 announcement in the Federal Register, 83 FR 3189.

²²DoL announced it would permanently switch to this new form of sequential processing in June 2018. ²³See, for example, https://nickarnosti.com/blog/h2bvisas/.

²⁴See the USCIS announcement on the allocation procedure, USCIS Completes Random Selection Process for H-2B Visa Cap for Second Half of FY 2018.

²⁵Authors' calculations from DoL Office of Foreign Labor Certification data, described in Section 4.

processed before February 27, giving early applicants a chance to proceed to the second step in the H-2B worker procurement process. Meanwhile, only 19% of TLC applications received after 7:00am on January 1 were processed before February 27. Thus, the majority of firms who sent in applications after 7:00am were excluded from the original tranche of 33,000 2HFY H-2B workers.

Figure 4: Proportion of 2HFY 2018 TLC Applications Processed before February 27, by Hour of Day Received on January 1



Source: DoL Office of Foreign Labor Certification Performance Data.

A second tranche of 15,000 "supplemental" visas was announced on May 25, 2018. The final rule was effective from May 31 through the end of the fiscal year. The process of allocating those visas was the same as for the initial visa allotment. As petitions once more exceeded the new allotment, USCIS conducted a second lottery, for these supplemental visas. The lottery results were announced June 11. Some firms who were originally unable to get 2nd Half FY2018 H-2B workers—either because their TLC applications were processed too late to participate in the first lottery, or because they lost the first lottery—had a second chance to do so. As we show in Section 6, this appears to have helped some firms that applied for TLC certifications after 7:00am on January 1, 2018.

3 Identification Strategy

Our identification strategy relies on the unexpected change in the ability to hire workers through the H-2B visa program, stemming from the January 1, 2018 TLC application spike. Based on the pattern shown in Figure 4, we define our exposure variable of interest as (Prop. Applications Before 7am)_j: the proportion of a firm j's January 1, 2018 TLC applications sent in before 7:00am.

While the criticality of getting applications in before 7:00am was not possible to foresee, there are still reasons to believe that firms applying before and after this cutoff might have differed. For example, lawyers submitting TLC applications on firm owners' behalf may prioritize certain clients. To address this concern, we estimate the following model specification:

$$y_{jt} = \sum_{\tau \neq 2017} \beta_{\tau} \left[(\text{Prop. Applications Before 7am})_j \times \mathbb{1}\{t = \tau\} \right] + \Gamma X_{jt} + \alpha_t + \alpha_j + \varepsilon_{jt} \quad (1)$$

where firm fixed effects, α_j , control for unobserved, time-invariant characteristics that correlate with firm application timing and subsequent firm performance. As such, our identifying assumption is that the outcomes of early-applicant firms submitting their TLC applications before 7:00am on January 1, 2018, would have trended similarly to those of later-applicant firms had it not been for the fact that the early-applicants had access to the original tranche of H-2B workers in the second half of FY 2018.²⁶ Estimates of β_{2015} and β_{2016} help assess the plausibility of this assumption.

When we include X_{jt} , this assumption becomes conditional on industry-by-year fixed effects, state-by-year fixed effects, and size-quartile-by-year fixed effects. Here, industry refers to a 6-digit NAICS code controlling for industry-wide labor demand shocks at a detailed level. Our state-by-year fixed effects capture geographic labor market trends, state-level policies, and other state-level characteristics.²⁷ In the case of multi-unit firms, state refers to the state in which a firm has the largest number of employees. Size is defined by the number of TLC applications the firm sent in on January 1, 2018. Therefore, size quartile indicators capture both firm size and reliance on the H-2B program.

Unless otherwise specified, outcome variables are divided by a common denominator: 2017 employment.²⁸ This helps remove skewness from our outcome variables while retaining the ability to compare coefficient estimates on a one-for-one basis (Roth and Chen, 2022).

²⁶This is a slight abuse of language since (Prop. Applications Before $(7am)_j$ is continuous, but it nonetheless conveys the intuition behind our identification. Furthermore, (Prop. Applications Before $(7am)_j$ is either 0 or 1 in the vast majority of cases. This is because firms pool applications for workers in the same occupation and worksite onto one request form, and the vast majority of firms only request workers in one occupation and one worksite.

²⁷Of particular note is time zone, given that firms located in states on Pacific Standard Time could have sent their TLC applications in at 9:00pm on December 31, 2017 and therefore may have had an advantage over firms in states on Eastern Standard Time, who had to wait until midnight. In practice, we do not see evidence for such an advantage: we find no correlation between time zone and (Prop. Applications Before 7am)_i.

²⁸For single-unit firms, 2017 employment is taken as the average across all four calendar quarters. For multi-unit firms, 2017 employment is measured only in calendar Q1.

4 Data

To examine how access to H-2B workers impacts firms, we combine administrative data on H-2B visa applications from the Department of labor and firm-level panel data compiled by the US Census Bureau.

We first gather data on firms' H-2B TLC applications from the Office of Foreign Labor Certification at the DoL.²⁹ These records include the universe of H-2B TLC Applications and the certification determinations from the Department's Office of Foreign Labor Certification, Employment, and Training Administration. For each submission, we have information on the employer name, address, state, ZIP code, the number of requested workers, the number of certified requests, and the date the application was submitted. Crucially, for FY 2018, the data includes the millisecond on January 1 at which applications were received, which allows us to construct (Prop. Applications Before 7 am)_i.

We also obtain data on i129 petitions from the USCIS.³⁰ The USCIS data is available for FY 2015 onward and contains information on employer name, address, state, and ZIP code, along with the number of approved i129 petitions for each employer.

To gauge the impact of access to H-2B workers on firm-level outcomes, we use the revised 2019 version of the Longitudinal Business Database (LBD) (Chow et al., 2021) granted to us by the US Census Bureau. The LBD is an establishment-level panel data set constructed from administrative tax records for each US non-farm, employee-hiring, private-sector establishment. Establishments are assigned unique, consistent identifiers that can be linked over time to create a true panel. It contains establishment-level information on calendar Q1 employment and annual payroll for all employers. Critically, it also contains establishment-level quarterly employment and payroll for single-establishment firms. Given that 2HFY H-2B visa workers are employed in calendar Q2 and Q3, this level of granularity helps us examine employer-level outcomes during the specific H-2B hiring period we study. Quarterly employment is measured on March 12, June 12, September 12, and December 12 for calendar Q1, Q2, Q3, and Q4, respectively.

The timing of measurement—in conjunction with the provision of the supplemental tranche of H-2B visas in the second half of FY 2018—impacts how we interpret our employment results. Given that the supplemental visa lottery results were announced on June 11, it is highly unlikely that H-2B workers on supplemental visas were on firm payrolls by June 12, but highly likely that they were on payrolls by September 12. Thus, when estimating how many total employees are hired per H-2B approval, we only consider the initial tranche of H-2B approvals for calendar Q2. In contrast, both initial and supplemental tranche H-2B approvals are taken into consideration in calendar Q3.

²⁹This data is publicly available from the Department of Labor Performance Data website.

³⁰This data is available from the USCIS H-2B Employer Data Hub.

The unique firm identifiers in the LBD allows us to link establishments to their parent firms. We thus collapse establishment-level variables to the firm level using these identifiers. A majority of firms are matched to annual revenues from the Census Bureau's BRFIRM_REV data set starting in 1997 (see Haltiwanger et al., 2019). Revenue data are only available through 2018.

Given that the H-2B data and the firm-level data (LBD) contain the employer name, state, city and ZIP code, we are able to link them using a fuzzy matching procedure.³¹ We match 90.2% of January 1, 2018, TLC applications to a source firm in the US Census Bureau data, resulting in a sample of roughly 3,200 firms that sent in TLC applications to the DoL on January 1, 2018.³² For most of the analyses, we focus on the subset of firms that were active in 2017 and 2018, but we then separately analyze the extensive margin outcome of whether or not a firm is actively operating in the full 3,200 firm sample.

4.1 Characteristics of H-2B Firms and Workers in the Research Sample

Table B2 presents the top five industries and occupations reported on TLC applications for H-2B workers on January 1, 2018. Landscaping is the dominant industry for 2HFY H-2B users, accounting for 52.6% of all TLC applications sent in on January 1, 2018 alone. Correspondingly, Landscaping and Groundskeeping is the dominant occupation for H-2B workers. The other key industry is Hotels and Motels (except casinos), which corresponds closely with the Maids and Housekeeping occupation. Generally, the occupations reported in Table B2 are associated with low-wage, manual work.

This notion is confirmed by the hourly wages associated with each request, which are reported in the DoL TLC application data, visualized in Figure 5. Reported hourly wages are tightly and roughly symmetrically distributed around a mean of \$13.24 2018 per hour.³³ About 80% of TLC applications feature hourly wages below \$15 per hour, and more than 99% feature hourly wages below \$22 per hour. Based on the data, H-2B visa workers generally make less than 60% of the median wage and salary US worker on a weekly basis.³⁴

A common characteristic of the H-2B industries is that they either sell directly to local customers or primarily provide inputs to firms that sell locally. Thus, the geography of applications is informative of the location of the impact of the H-2B visa program. Figure 6 shows that this impact is likely to be geographically dispersed.

³¹See Appendix A for a description of the process.

³²All firm counts are rounded in accordance with Census Bureau requirements.

³³Standard deviation: \$2.44; Median: \$13.29

³⁴We multiplied weekly hours (also reported on TLC applications) with hourly wages to arrive at a weekly wage, which we then compared to the median weekly wages for all workers aged 16+ from the Bureau of Labor Statistics "Weekly and hourly earnings data from the Current Population Survey" release for 2018 Q2.

Figure 5: Distribution of Hourly Wages Reported on Certified Jan. 1, 2018 TLCs



Notes: In limited cases ($\approx 1\%$) in which wages are reported on a weekly basis, they are divided by expected weekly hours—which are also reported in TLC applications—to generate an hourly wage. Wages are in nominal terms, to be paid in calendar Q2 and Q3 of 2018. **Source:** DoL Office of Foreign Labor Certification Performance Data.

Figure 6: Geographic Distribution of January 1, 2018 TLC Applications



Notes: Numbers correspond to raw counts of TLC applications based on employer location (county). Figure B1 re-plots these counts as a fraction of county employment. **Source:** Department of Labor Office of Foreign Labor Certification Performance Data

Finally, Table 1 provides some context for our sample by comparing the firms in the dataset to others in the US economy. Firms applying for H-2B workers tend to be larger across several dimensions. For instance, the median firm in our sample grossed \$2.1 million in revenues, compared to \$1 million in the case of the median US firm. In terms of employment, the median H-2B employer in our sample is approximately three times as large as the average US employer. These characteristics are consistent with Mahajan (2022), who finds that H-2B users are concentrated toward the top of the labor productivity distribution. Thus, despite their concentration in a few sectors, these firms have the potential to play a significant role in their local economies.

		Research Sample (H-2B Firms)			US Economy		
Variable	Universe	Mean	Std. Dev.	Median^{\S}	Mean	Median	
Revenues $(\$1,000)$	All Private Sector	$6,\!634$	$54,\!516$	$2,\!101$	$6{,}232^{\dagger}$	$[500, 1, 000)^{\dagger}$	
Employment	All Private Sector	34.6	69.0	17.8	21.44^\dagger	$[1,5)^{\dagger}$	
2HFY Employment	Single-Unit Only	41.7	79.0	22.27	$10.21^{\dagger\dagger}$	$[3,9]^{\dagger\dagger}$	
Payroll (\$1,000)	All Private Sector	1,404	$2,\!665$	713.7	$1{,}121^\dagger$	$\approx 211^{\dagger\dagger\dagger}$	

Table 1: 2017 Firm-Level Summary Statistics

Notes: Source for all statistics in "Research Sample" is the Longitudinal Business Database and associated linked data sets from the US Census Bureau. Employment refers to mean employment across all four quarters for single-unit firms and Q1 employment for multi-unit firms in the research sample. 2HFY Employment refers to mean employment across the second and third quarters in a calendar year (the second half of the fiscal year) for single-unit firms. Monetary values are in 2017 USD, deflated by the GDP Implicit Price Deflator series.

[§] For the research sample, reported medians are means taken between the 40th and 60th percentile in accordance with Census Bureau requirements.

 † Source: Statistics of US Businesses, Census Bureau—firm counts by revenue bins.

^{††} Source: Business Dynamics Statistics—Single-Unit (BDS-SU), Census Bureau—firm counts by employment bins. ^{†††} Source: Statistics of US Businesses, Census Bureau—mean payroll among firms with \$500-1,000 thousand in revenues.

5 Application Timing and H-2B Outcomes

5.1 Additional Support for Identification Assumptions

We start by presenting evidence that firms were unaware of the importance of sending their applications to the DoL by 7:00am on January 1, 2018. If firms had anticipated the application spike, one would expect those with higher demand for H-2B workers to have submitted their applications earlier. In addition, expecting increased competition for the limited number of provided visas, one would expect these firms to send in higherquality applications with a higher likelihood of being certified. To assess any possible anticipation, we estimate a placebo regression using TLC applications as a measure of demand in Equation (1). To proxy application quality, we use the final number of TLC certifications by the DoL.³⁵ Figure 7 presents the estimated coefficients $\hat{\beta}_{\tau}$.

The left panel of Figure 7 does not present evidence of a difference in the number of TLC applications between early- and late-applicants *through* 2018. Our estimates in 2018 are small and not statistically different from zero at conventional significance levels.

³⁵Recall that, conditional on application quality, early-applicants are no less more to be certified than late applicants; rather, they are more likely to be certified *on time*.

The results are virtually unchanged when we do not include any control variables.³⁶ We also obtain similar results when using our proxy for application quality in the right panel of Figure 7.

In sum, both the number and likely quality of applications were not significantly different for early- and late-applicants on January 1, 2018, suggesting they were unaware of the changes in processing that DoL would announce ex-post, on January 17, 2018.



Notes: See Equation (1) and Section 3 for details of specification. Capped spikes around coefficient estimates represent 95% confidence intervals, generated from standard errors that are clustered at the firm level. Each specification estimated on 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects and size-by-year fixed effects. Outcome numerators count the total number of TLC applications filed by a firm (left panel) and total number of TLC applications that were certified for a firm in a given fiscal year (right panel). All outcomes are divided by 2017 employment at the firm.

5.2 Effect of Application Timing on Eventual H-2B Visa Approvals

We next establish that TLC application submission prior to 7:00am on January 1, 2018 was a strong predictor for the likelihood of getting an H-2B worker petition approved by USCIS for the 2nd half of the fiscal year. To do so, we use the data on i129 petitions from the USCIS, which distinguishes between FY 1st Half, original tranche 2HFY, and supplemental 2HFY visa petitions, along with whether petitions are for initial employment or continuing workers. This enables us to conduct a precise test of the "first stage:" the effect of (Prop. Applications Before 7am)_j on firms' ability to receive approvals for *original tranche*, 2HFY i129 petitions for initial employment in 2018.

Figure 8 plots the coefficients $\hat{\beta}_{\tau}$ from estimating Equation (1) using USCIS visa petition approvals as the outcome. The left panel uses approved i129 petitions subject to the original fiscal year 2HFY cap (Original Tranche 2HFY). In the right panel, we

³⁶Given that the size-bin-by-year fixed effects are based on the number of January 1, 2018 TLC applications, one may worry that we are conditioning on the outcome in the "TLC Applications" panel of Figure 7. This finding does not change with the exclusion of controls (light blue circles).

also include approved i129 petitions subject to the 2HFY supplemental cap and approved 2HFY i129 petitions for continuing employment (All 2HFY).³⁷



Figure 8: The Effect of Application Timing on H-2B Approvals

Notes: See Equation (1) and Section 3 for details of specification. Capped spikes around coefficient estimates represent 95% confidence intervals, generated from standard errors that are clustered at the firm level. Each specification estimated on 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects, and size-by-year fixed effects. The outcome numerator in the left panel counts the total number of initial USCIS H-2B approvals for work in the second half of a fiscal year for a given firm. Outcome numerator in right panel counts the total number of initial USCIS H-2B approvals for work in the second half of a fiscal year for a given firm: initial, continuing, and supplemental. All outcomes are divided by 2017 employment at the firm.

Two apparent features of the left panel of Figure 8 further support the validity of our research design. First, we do not find evidence of H-2B usage trending any differently for earlier-applicants when compared to later-applicants prior to 2018. Second, we estimate a substantive increase in H-2B visa approvals in the second half of FY 2018 for early applicants. The results in the left panel imply that early applicants obtained 11.2 additional H-2B original tranche visa approvals for the second half of FY 2018.³⁸ These results support the notion that application timing created unexpected luck in the 2HFY H-2B procurement process in 2018.

In the right panel of Figure 8, we re-run the event study including the supplemental visas in the outcome variable. The results indicate that the supplemental visas may have helped late-applicants recoup some of their ability to hire H-2B workers. When we account for the supplemental tranche of visas, the effect of applying for TLCs prior to 7:00am on January 1 on receiving *any* H-2B visa approvals for the second half of FY 2018 is about 70% as big as the effect on receiving H-2B visa approvals from the original tranche. Nevertheless, even with the additional supplemental visas, firms applying after 7:00am were left with approximately 7.7 fewer H-2B visa approvals relative to early-applicants.³⁹

³⁷Continuing employment visas are not cap subject but were exceedingly rare in 2018 and therefore do not play a role in these results.

 $^{^{38}}$ Multiplying the 2018 coefficient estimate by the mean of 2017 firm employment.

³⁹This time multiplying the 2018 coefficient in the right panel with the mean of 2017 firm employment.

6 Results

6.1 Does H-2B Hiring Crowd Out Other Employment?

We start with the canonical question of whether the hiring of foreign-born workers crowds out other forms of employment by looking directly at how access to H-2B workers affect firms' total employment counts. This assessment hinges crucially on the fact that H-2B workers are subject to federal income taxes and, therefore, enumerated in the firm-level employment counts. As we observe all workers subject to federal income taxes in our data, each H-2B hire should add one employee to the firm during the hiring period corresponding to calendars Q2 and Q3 in 2018 in the absence of crowd out. Conversely, if there is crowd out (crowd in), we would expect firm's employment to increase by less than one (more than one).

In Figure 9, we plot the results of estimating Equation (1) with quarterly employment as outcome. Being eligible to use the H-2B program has no impact on Q1 employment, as shown in the first upper panel in the figure. As Q1 employment is a count of employees at the firm on March 12 and 2HFY H-2B workers cannot start work before April 1, the null effect is reassuring, as it serves as an additional placebo check for anticipatory effects.

In calendar Q2, when firms are eligible to employ H-2B workers, we find a substantive increase in employment (upper right panel of Figure 9). Dividing the coefficient by the corresponding original tranche i129 approval estimates in the left panel of Figure 8 yields an estimate of 0.64 employees per H-2B *approval.*⁴⁰ We view this estimate as a lower bound on the number of additional employees per H-2B *hire*. Since roughly 78% of approved H-2B petitions result in an actual hire, our estimates suggest an increase of around 0.82 employees per H-2B hire.⁴¹ This effect is slightly less than a one-to-one increase in employment per H-2B hire, suggesting extant but limited crowd-out of other workers.

These findings are in line with those from Clemens and Lewis (2022), who do not find any significant changes in the hiring of US workers by firms participating in the H-2B program, so long as the small crowd-out suggested above is of *other* (non-H-2B) temporary, foreign-born workers. Our results contrast strongly with those in Doran et al. (2022), who can not reject full crowd-out of other employees in response to H-1B hiring. Given that H-1B workers are required to have college degrees and earn substantially more than H-2B workers, we highlight the possibility that the hiring of foreign-born workers

⁴⁰Recall from Section 4 that workers on supplemental H-2B visas had not started working for employers by June 12, 2018 when Q2 employment was measured for 2018.

 $^{^{41}}$ See our discussion in Section 2.2.

may have substantively different implications for incumbent workers based on where they stand in the wage distribution.

We can conduct a similar exercise for Q3 employment, but using *all* 2HFY H-2B visa approvals as the denominator to account for the supplemental visas made available and likely realized before September 12, 2018. As above, we estimate that each 2HFY H-2B visa approval leads to 0.62 additional employees per approval in Q3. Using our benchmark 78% H-2B approval-to-hire conversation rate, this suggests an increase of 0.80 in Q3 employees per H-2B hire. This effect is very close to the estimated impact for Q2, suggesting firms kept a relatively stable workforce over the period.

Finally, the impact of H-2B hiring does not extend to Q4 employment (lower right panel of Figure 9). The estimated coefficient is close to zero and not statistically different from zero. This result implies firms eligible to hire through the program did not significantly alter the size of their workforce in the long run. In fact, the estimated employment impacts after 2018Q3 are all close to zero. In sum, the employment results suggest that access to the H-2B program helped firms address temporary labor needs.



Figure 9: The Effect of Application Timing on Employment Counts, by Quarter

Notes: See Equation (1) and Section 3 for details of specification. Each specification estimated on 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects and size-by-year fixed effects. Outcome numerators are quarterly employment counts, measured on the indicated dates. All outcomes are divided by 2017 employment at the firm.

6.2 Effects on Payrolls and Average Pay

Critics of the H-2B visa program have long been concerned about how its impact on incumbent workers' wages. To assess the validity of such concerns, Figure 10 plots the estimated coefficients from Equation (1) using 2HFY payroll per worker as the outcome.⁴² The results do not support the notion that firms lower worker pay when given access to the H-2B program. We also do not find evidence of payroll per worker trending any differently prior to 2018 for early- vs. late-applicants.



Figure 10: The Effect of Application Timing on Pay per Worker

Notes: See Equation (1) and Section 3 for details of specification. Capped spikes around coefficient estimates represent 95% confidence intervals, generated from standard errors that are clustered at the firm level. Each specification estimated on 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects and size-by-year fixed effects. Outcome in left panel is the log of total Q2 and Q3 payroll divided by mean Q2 and Q3 employment. All monetary variables are in real 2009 USD, deflated by the GDP Implicit Price Deflator series.

One possibility, however, is that effects on average pay mask changes to within-firm pay dispersion. This could occur if, for example, firms save on labor costs using H-2B workers while raising the compensation of higher-paid incumbent workers. Unfortunately, we cannot assess if that is the case because we do not have data on individual compensation. However, we can gauge if changes in payroll appear consistent with the one-to-one increase in H-2B employment, roughly matching the required compensation for an H-2B worker.

Figure 11 plots the impact of early application on quarterly payrolls. As with the employment results, we do not find evidence of a significant impact in Q1, when 2HFY H-2B workers were not yet on firm payrolls (upper left panel). In contrast, accompanying the increase in firms' employment documented above, early-applicants' payrolls rise by \$1,439 in Q2 \$1,332 in Q3 in 2018.

 $^{^{42}}$ The denominator is the mean of Q2 and Q3 employment.



Figure 11: The Effect of Application Timing on Payroll, by Quarter

Notes: See Equation (1) and Section 3 for details of specification. Each specification estimated on 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects and size-by-year fixed effects. Outcome numerators are quarterly payroll, in thousands of real 2009 USD, deflated by the GDP Implicit Price Deflator series. All outcomes are divided by 2017 employment at the firm.

Using the same conversion rates as we do above, these figures translate to \$4,458 per H-2B approval in Q2 and \$5,960 per H-2B approval in Q3, and \$5,715 per H-2B hire in Q2 and \$7,631 per H-2B hire in Q3. Meanwhile, the average weekly wage rate for an H-2B worker is around \$450. Given that a fiscal quarter has 13 weeks, a full-time H-2B worker should earn roughly \$5,800 per quarter. Thus, our results are not supportive of the notion that firms use the H-2B visa program to undercut wages. Instead, we find evidence for positive spillovers on non-H-2B workers in Q3.

The small but statistically significant payroll increases seen among early applicant firms in Q4 of 2018 may also be the result of spillovers, given that employment effects have abated by Q4 in Figure 9. However, they may also be the result of the fact that payroll is measured for the whole quarter, whereas employment is only measured on December 12. While the majority (71%) of 2HFY H-2B applications indicate an end date before December 12, a majority also indicate an end date after October 1 (98%).

In sum, the results on employment and payroll indicate that the scope for crowd-out of existing employment is rather limited. Instead, they support the notion of firms' usage of the H-2B program to address temporary labor needs, with limited impacts on other employment and the potential for positive pay spillovers to non-H-2B workers. However, we caution that 1) these are estimates of firm impacts on the intensive margin, among operating firms, that do not incorporate firms that ceased operation; and 2) the estimated 78% conversion rate from H-2B approval to H-2B hire is taken from USCIS reports for FY 2007–2015 and is an approximation of the FY 2018 conversion rate. We separately examine extensive margin impacts in Section 6.3.

6.3 Does H-2B Hiring Affect Firm Performance?

Thus far, we have not found any evidence of access to H-2B workers crowding out other forms of employment or lowering workers' pay. Next, we investigate the impact that H-2B hiring might have on firm performance by focusing on revenues and firm survival.

Revenues

We first turn to the impact of H-2B hiring on firms' annual revenues.⁴³ If firms rely on the H-2B program to address hiring constraints, lifting that constraint should result in higher revenues.



Figure 12: The Effect of Application Timing on Annual Revenues

Notes: See Equation (1) and Section 3 for details of specification. Capped spikes around coefficient estimates represent 95% confidence intervals, generated from standard errors that are clustered at the firm level. Each specification estimated on 2,400 firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with observed, positive annual revenues in 2017 and 2018. Observations weighted by inverse probability weights designed to make revenue sample consistent with the universe of private sector firms. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects and size-by-year fixed effects. Outcome in left panel is annual revenues divided by 2017 mean employment. Outcome in right panel is the log of annual revenues divided by mean quarterly employment in a given year. Numerator is in thousands of real 2009 USD, deflated by the GDP Implicit Price Deflator series.

⁴³Recall that annual revenues are available for subset of firms, but that this subset *does* include multiestablishment firms, unlike our quarterly payroll and employment results.

The left panel in Figure 12 reveals that this is the case. Early applicants earn an additional \$9.5 thousand in revenues in 2018 relative to late applicants, which roughly translates to an additional \$42 thousand in revenues per H-2B approval and \$54 thousand per H-2B hire.⁴⁴ This finding is also in line with the results in Clemens and Lewis (2022), who document a production expansion among firms winning the H-2B lottery. We also note that in a wide swath of theoretical models that feature imperfect competition in the product market (e.g., Dixit and Stiglitz, 1977), revenue increases also imply output price decreases. In the context of such models, our results lend some credence to the view that an expansion of the H-2B program may help lower price levels.

We do not find evidence of increases in per worker annual revenues (right panel of Figure 12). In conjunction with our results on payroll per worker, this suggests firms do not extract additional rents from hiring temporary H-2B workers.⁴⁵

Firm Survival

Firms applying for H-2B workers often state that access to these temporary labor is vital to their survival. We use an indicator for whether or not a firm reports positive annual payroll to assess such claims. Our estimates from Equation (1) using this indicator as an outcome are presented in Figure 13.





Notes: See Equation (1) and Section 3 for details of specification. Capped spikes around coefficient estimates represent 95% confidence intervals, generated from standard errors that are clustered at the firm level. Each specification estimated on 3,200 firms that applied to the Department of Labor for a TLC certification on January 1. Specifications with controls include industry-by-year fixed effects state-by-year fixed effects and size-by-year fixed effects. Outcome is an indicator of whether a firm had positive payroll in a given year.

⁴⁴The scaling for approvals comes from dividing $\hat{\beta}_{2018}$ in the left panel of Figure 12 by $\hat{\beta}_{2018}$ from the right panel of Figure 8. However, we note that the underlying samples are different in the two figures.

⁴⁵One would ideally observe profits to evaluate rent-sharing. Profits are, however, not available in our data. In line with our interpretation, Clemens and Lewis (2022) find that being eligible for using the H-2B program does not change a firm's profit rate.

The likelihood of remaining active is 1.59 percentage points higher among early applicants in 2018, and this effect rises to 3.06 percentage points by 2019. One potential explanation could be that firms (correctly) saw the 2018 application spike as a harbinger for increased difficulty in obtaining H-2B workers in the future, so that those that had previously relied on this workforce for survival no longer forecast long term profitability.

Our results support the notion that low-wage migrant workers are essential to survival among a subset of firms. To put them in perspective, when assessing the impact of import shocks from low-wage countries on U.S. firms, Bernard et al. (2006) estimate that a one standard deviation increase in import competition from low-wage countries decreases firm survival in the same industry and within the next five years by 2.2 percentage points. Thus, our estimates indicate that the inability of firms to hire H-2B workers in the very short term has a similar impact on firm survival as facing longer-run import competition from low-wage countries.

7 Heterogeneity and Spillovers Analysis

Access to H-2B workers may have differential impacts across firms and markets on various dimensions. To learn about heterogeneous effects, we consider the following model specification:

$$\Delta y_j = \beta (\text{Prop. Applications Before 7 am})_j + \gamma Z_j + \delta \left[(\text{Prop. Applications Before 7 am})_j \times Z_j \right] + \Gamma X_j + \varepsilon_j.$$
(2)

and focus on three outcomes meant to summarize the main findings from Section 6 — namely, the 2017-2018 change in 2HFY employment (mean calendar Q2 and Q3 employment), the 2017-2018 change in annual revenues, and the 2017-2019 change in the activity indicator.⁴⁶ Z_j represents one of the stratifying variables described in the subsections below. Estimated coefficients from specifications without Z_j are also presented for comparison.

7.1 Heterogeneity by Firm Productivity Measures

We first examine how our findings vary based on two proxies of firm productivity in the base period—namely, the firm's 2017 standardized log revenues and its 2017 standardized log revenues per worker. Because these stratifying variables are standardized, δ measures

⁴⁶Without the terms for Z_j and [(Prop. Applications Before 7 am)_j × Z_j], the estimated coefficient indicated by β exactly corresponds to the β_{2018} coefficient from Equation (1) when 2HFY employment and revenues are the outcomes. It also corresponds to the β_{2019} coefficient from Equation (1) when the firm activity indicator is the outcome.

how much the treatment effect captured by β changes when productivity levels rise by one standard deviation above the mean.

Based on the results displayed in Table 2, we do not find evidence of access to H-2B workers having a greater impact on the employment or revenues of initially larger firms. Nevertheless, relative to smaller firms, larger early-applicant firms exhibit a higher survival likelihood a year later. Specifically, a one standard deviation increase in firms' initial revenues raises the likelihood of survival of early applicants by 5 percentage points.

Outcome:	2HFY Employment		Annual Revenues		Activity Indicator	
	(1)	(2)	(3)	(4)	(5)	(6)
Prop. Applications Before 7 am_j	$\begin{array}{c} 0.173^{***} \\ (0.024) \end{array}$	$\begin{array}{c} 0.172^{***} \\ (0.0241) \end{array}$	9.49^{**} (4.31)	9.596^{**} (4.454)	$\begin{array}{c} 0.031^{***} \\ (0.010) \end{array}$	0.030^{***} (0.0099)
Std. Log 2017 $\operatorname{Revenues}_j$		-0.006 (0.030)		-4.54 (3.04)		-0.002 (0.010)
Prop. Applications Before 7 am_j \times Std. Log 2017 $\mathrm{Revenues}_j$		-0.023 (0.034)		$4.90 \\ (3.47)$		0.020^{*} (0.011)
Firms	2,800		2,400		3,200	
Change Taken Over	2017-2018		2017-2018		2017-2019	

Table 2:	Heterogeneous	Effects	bv	Initial	Firm	Size
	0		•/			

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. In Columns (1) and (2), the outcome variable is the 2017-2018 change in 2HFY employment divided by 2017 mean employment. Columns (1) and (2) are estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. In Columns (3) and (4), the outcome variable is the 2017-2018 change in annual revenues divided by 2017 mean employment. Columns (3) and (4) are estimated on the set of 2,400 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive annual revenues in 2017 and 2018. In Columns (5) and (6), the outcome variable measures the 2017-2019 change in an indicator for whether or not a firm has positive payroll. Columns (5) and (6) are estimated on the set of 3,200 firms that applied to the Department of Labor for a TLC certification on January 1, 2018. All specifications include firm fixed effects, state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

Access to H-2B workers appears to have a stronger impact on initially more productive firms as well. Based on the results in Column (2) of Table 3, a one standard deviation increase above the mean in firms' initial revenues per worker increases the employment impacts of access to H-2B workers. This impact also translates into higher annual revenues (i.e., Column (4) with p = 0.11) and a higher likelihood of firm survival through 2019 (see Column (6)).

Overall, the results in Tables 2 and 3 are suggestive of larger and more productive firms benefiting the most from the H-2B program, especially with regards to firm survival throughout 2019. This finding is in line with prior studies showing how lower-wage immigrant workers may prove critical to business survival among higher-productivity and larger businesses (Mahajan, 2022; Ayromloo et al., 2020).

Outcome:	2HFY Employment		Annual Revenues		Activity Indicator	
	(1)	(2)	(3)	(4)	(5)	(6)
Prop. Applications Before 7 am_j	$\begin{array}{c} 0.173^{***} \\ (0.024) \end{array}$	$\begin{array}{c} 0.173^{***} \\ (0.0242) \end{array}$	9.49^{**} (4.31)	10.80^{**} (4.96)	$\begin{array}{c} 0.031^{***} \\ (0.010) \end{array}$	$\begin{array}{c} 0.031^{***} \\ (0.010) \end{array}$
Std. Log 2017 Rev. pw_j		0.046^{*} (0.025)		-23.80 (14.96)		-0.015 (0.012)
Prop. Applications Before 7 am_j \times Std. Log 2017 Rev. pw_j		0.099^{**} (0.041)		$23.95 \\ (15.15)$		0.023^{*} (0.013)
Firms Change Taken Over	2,800 2017-2018		2,400 2017-2018		3,200 2017-2019	

Table 3: Heterogeneous Effects by Initial Firm Labor Productivity

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. In Columns (1) and (2), the outcome variable is the 2017-2018 change in 2HFY employment divided by 2017 mean employment. Columns (1) and (2) are estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. In Columns (3) and (4), the outcome variable is the 2017-2018 change in annual revenues divided by 2017 mean employment. Columns (3) and (4) are estimated on the set of 2,400 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive annual revenues in 2017 and 2018. In Columns (5) and (6), the outcome variable measures the 2017-2019 change in an indicator for whether or not a firm has positive payroll. Columns (5) and (6) are estimated on the set of 3,200 firms that applied to the Department of Labor for a TLC certification on January 1, 2018. All specifications include firm fixed effects, state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

7.2 Heterogeneity by Immigration Enforcement Environment

Given that the purpose of the H-2B program is to increase employers' access to foreign workers to address temporary labor needs, its usage may vary with interior immigration enforcement policies that restrict firm access to unauthorized workers. Because unauthorized workers are more likely to be less educated and occupy low-wage jobs, they may have much in common with foreign-born low-wage workers hired through the H-2B program (Passel and Cohn, 2009). Firms exposed to greater interior immigration enforcement measures may have been more prone to turn to the H-2B program in order to meet labor needs.

To assess how the policy environment may have impacted firms' response to the H-2B program, we focus on the effect of employment-based immigration enforcement as exemplified by employment verification (E-Verify) mandates. E-Verify mandates require firms to check the work eligibility of prospective employees using an internet-based program that compares information entered by an employer from the I-9 form against information from the Department of Homeland (DHS) and Social Security Administration (SSA) databases. Universal mandates—which require all public and private employers to

use E-Verify—have been adopted by Alabama, Arizona, Georgia, Louisiana, Mississippi, North Carolina, South Carolina, Tennessee, and Utah.⁴⁷

For this analysis, we set Z_j in Equation (2) equal to one when firm j's state of main operation, s(j)—defined as the state in which the firm employed the most workers—has a universal E-Verify mandate in place. Otherwise, it is set equal to zero. Results from this heterogeneous analysis are displayed in Table 4. Overall, the findings underscore the relevance of increased access to H-2B workers on firm survival, and show that these effects are magnified in E-Verify states. This result suggests that H-2B and unauthorized immigrant workers may be substitutes in production and speaks to the broader literature on the interaction between enforcement policy and legal channels of U.S. migration (Orrenius and Zavodny, 2020; Amuedo-Dorantes et al., 2021). While we also find a positive effect at the intensive margin, we lack the precision to conclude if it is statistically significant.

Outcome:	2HFY Employment		Annual Revenues		Activity Indicator	
	(1)	(2)	(3)	(4)	(5)	(6)
Prop. Applications Before 7 am_j	$\begin{array}{c} 0.173^{***} \\ (0.024) \end{array}$	$\begin{array}{c} 0.161^{***} \\ (0.026) \end{array}$	9.49^{**} (4.31)	5.31^{**} (2.64)	$\begin{array}{c} 0.031^{***} \\ (0.010) \end{array}$	0.022^{**} (0.010)
Prop. Applications Before 7 am_j \times E-Verify_{s(j)}		$0.095 \\ (0.087)$		33.17 (27.65)		0.065^{**} (0.033)
Firms	2,800		2,400		3,200	
Change Taken Over	2017-2018		2017-2018		2017-2019	

Table 4: Heterogeneous Effects by Firm Exposure to E-Verify Mandates

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. In Columns (1) and (2), the outcome variable is the 2017-2018 change in 2HFY employment divided by 2017 mean employment. Columns (1) and (2) are estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. In Columns (3) and (4), the outcome variable is the 2017-2018 change in annual revenues divided by 2017 mean employment. Columns (3) and (4) are estimated on the set of 2,400 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll change in an indicator for whether or not a firm has positive payroll. Columns (5) and (6) are estimated on the set of 3,200 firms that applied to the Department of Labor for a TLC certification on January 1, 2018. All specifications include firm fixed effects, state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects. State-by-year fixed effects absorb main effect of E-Verify_{s(j)}.

⁴⁷Prior studies have gauged how alternative immigration enforcement programs. Of particular note, East et al. (2023) find that Secure Communities—a police-based immigration enforcement program implemented by local or state-level police—significantly impacted local labor markets. However, Secure Communities was in place in all localities by 2014, prior to our study period. As such, we lack the much needed policy variation to examine the interaction between the H-2B program and Secure Communities.

7.3 Testing for Spillovers

Finally, we use Equation (2) to gauge if violations of the no-spillover assumption implicit in our analysis could be biasing our findings. To that end, we construct a "market-level" on-time application rate and contrast it with the firm-level on-time application rate that we have been using thus far. The market level rate is defined as:

Prop. Applications Before 7 am_{m(j)} = $\frac{\sum_{j' \neq j, \in m(j)} \text{Jan. 1, 2018 TLC Applications Filed Before 7:00am_{j'}}{\sum_{j' \neq j, j' \in m(j)} \text{Total Jan. 1, 2018 TLC Applications}_{j'}},$

to capture the proportion of January 1, 2018 TLC applications filed by firms, other than firm j in j's market, before 7:00am. Firm j's market, m(j), is defined by the cross of firm j's commuting zone of primary operation (based on employment) and 4-digit NAICS industry.

We then use the Prop. Applications Before 7 $\operatorname{am}_{m(j)}$ as the vector Z_j in Equation (2). In this case, γ can be interpreted as a measure of the spillover effect on firms that were excluded from the original tranche of 2HFY H-2B visas in 2018, but faced competition from firms that were not. δ can be interpreted as the portion of the treatment effect that is siphoned away by competitors' access to H-2B workers.

Table 5 presents the results of this exercise. Generally, our estimates of γ and δ in Equation (2) are not statistically significant, but point estimates are large. For example, in Column (4), we cannot reject the null hypothesis that market-level on-time application rates have no effect on either on-time or late applicants' revenues, but we also cannot reject the null hypothesis that all of a firm's revenue gains from on-time applications are siphoned away if all of their within-market competitors also applied on time. Thus, despite the lack of statistical significance presented in Table 5, we hesitate to draw strong conclusions about the existence of spillover effects. We do, however, note that the relative stability of our $\hat{\beta}$ estimates—across models in which we do and do not include Prop. Applications Before 7 am_{m(j)}—suggest that spillover effects are not generating substantial bias in our main results.

8 Conclusions

The impact that foreign-born workers who are willing to work in low-wage jobs have on the economy has been the center of numerous policy debates in advanced economies. Recently, these discussions have intensified due to growing labor shortages in sectors that typically employ low-wage workers. Despite these fervent debates, there is little well-identified evidence on how low-wage, foreign-born workers impact the firms that hire

Outcome:	2HFY Employment		Annual Revenues		Activity Indicator		
	(1)	(2)	(3)	(4)	(5)	(6)	
Prop. Applications Before 7 am_j	$\begin{array}{c} 0.173^{***} \\ (0.024) \end{array}$	$\begin{array}{c} 0.125^{***} \\ (0.026) \end{array}$	9.49^{**} (4.31)	$11.37^{**} \\ (5.72)$	$\begin{array}{c} 0.031^{***} \\ (0.010) \end{array}$	0.036^{**} (0.015)	
Prop. Applications Before 7 $\mathrm{am}_{m(j)}$		-0.061 (0.055)		$11.11 \\ (9.44)$		$0.009 \\ (0.022)$	
Prop. Applications Before 7 am_j \times Prop. Applications Before 7 $\operatorname{am}_{m(j)}$		$0.088 \\ (0.059)$		-9.18 (7.96)		-0.011 (0.023)	
Firms	2,8	800	2,	400	3,2	00	
Change Taken Over	2017	2017-2018		2017-2018		2017-2019	

Table 5: Testing for Spillovers

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. In Columns (1) and (2), the outcome variable is the 2017-2018 change in 2HFY employment divided by 2017 mean employment. Columns (1) and (2) are estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive payroll and employment in Q2 and Q3 of 2017 and 2018. In Columns (3) and (4), the outcome variable is the 2017-2018 change in annual revenues divided by 2017 mean employment. Columns (3) and (4) are estimated on the set of 2,400 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and with positive annual revenues in 2017 and 2018. In Columns (5) and (6), the outcome variable measures the 2017-2019 change in an indicator for whether or not a firm has positive payroll. Columns (5) and (6) are estimated on the set of 3,200 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018. All specifications include firm fixed effects, state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

them. Ultimately, the aggregation of these firm-level effects largely determines the overall impact of low-wage, foreign-born workers on the economy as a whole.

In this context, the H-2B visa program plays a critical role for a subset of US firms, offering them a channel through which to address their temporary labor needs. Yet, the program has been criticized as a potential means for firms to cut down on labor costs at the expense of native workers. Despite such claims, surprisingly little is known about the program's impact on workers and firms.

We exploited an unanticipated change in the processing of H-2B worker requests in 2018 that affected firms' ability to hire low-wage, foreign-born workers on a temporary basis. Using the near-universe of firms applying for the H-2B program, we find limited evidence of access to H-2B workers crowding out other forms of employment or adversely changing average pay per worker. Additionally, firms gaining access to H-2B workers appear to experience an increase in annual revenues and survival likelihood, suggesting the program may be helping firms address labor shortages.

In sum, our results point to the relevance of the H-2B program, in particular, and foreign-born workers, more generally, in alleviating firm-level hiring constraints. In addition, they challenge prevailing beliefs about the impact of low-wage, temporary migrants on employment and earnings, providing a different perspective on how access to temporary foreign labor affects workers and firms.

References

- Abramitzky, Ran, Philipp Ager, Leah Boustan, Elior Cohen, and Casper W. Hansen, "Local Labor Markets: Lessons from the 1920s Border Closure," *American Economic Journal: Applied Economics*, 2022.
- Amuedo-Dorantes, Catalina, Esther Arenas-Arroyo, and Bernhard Schmidpeter, "Immigration Enforcement and the Hiring of Low-Skilled Labor," *AEA Papers and Proceedings*, 2021, 111, 593–597.
- Ayromloo, Shalise, Benjamin Feigenberg, and Darren Lubotsky, "States Taking the Reins? Employment Verification Requirements and Local Labor Market Outcomes," NBER Working Papers 26676, National Bureau of Economic Research, Inc January 2020.
- Beerli, Andreas, Jan Ruffner, Michael Siegenthaler, and Giovanni Peri, "The Abolition of Immigration Restrictions and the Performance of Firms and Workers: Evidence from Switzerland," *American Economic Review*, 2021.
- Bernard, Andrew B., J. Bradford Jensen, and Peter K. Schott, "Survival of the Best Fit: Exposure to Low-Wage Countries and the (Uneven) Growth of U.S. Manufacturing Plants," *Journal of International Economics*, 2006, 68 (1), 219–237.
- Borjas, George J., "The Labor Demand Curve is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market," *Quarterly Journal of Economics*, 2003, 118 (4), 1335– 1374.
- Brinatti, Agostina and Nicolas Morales, "Firm Heterogeneity and the Impact of Immigration: Evidence from German Establishments," 2021. Working Paper.
- __, Mingyu Chen, Parag Mahajan, Nicolas Morales, and Kevin Shih, "The Impact of Immigration on Firms and Workers: Insights from the H-1B Lottery," Technical Report, SSRN 2023.
- Chow, Melissa C, Teresa C Fort, Christopher Goetz, Nathan Goldschlag, James Lawrence, Elisabeth Ruth Perlman, Martha Stinson, and T. Kirk White, "Redesigning the Longitudinal Business Database," Working Paper 28839, National Bureau of Economic Research May 2021.
- Clemens, Michael A. and Ethan G. Lewis, "The Effect of Low-Skill Immigration Restrictions on US Firms and Workers: Evidence from a Randomized Lottery," NBER Working Paper 30589, NBER 2022.
- _ , _ , and Hannah M. Postel, "Immigration Restrictions as Active Labor Market Policy: Evidence from the Mexican Bracero Exclusion," *American Economic Review*, 2018, 108 (6), 1468–87.
- Dixit, Avinash K. and Joseph E. Stiglitz, "Monopolistic Competition and Optimum Product Diversity," The American Economic Review, 1977, 67 (3), 297–308.
- **Doran, Kirk, Alexander Gelber, and Adam Isen**, "The Effects of High-Skilled Immigration Policy on Firms: Evidence from Visa Lotteries," *Journal of Political Economy*, 2022, 130 (10), 2501–2533.
- **Dustmann, Christian and Albrecht Glitz**, "How Do Industries and Firms Respond to Changes in Local Labor Supply?," *Journal of Labor Economics*, 2015, 33 (3), 711–750.

- __, Uta Schön berg, and Jan Stuhler, "Labor Supply Shocks, Native Wages, and the Adjustment of Local Employment," *Quarterly Journal of Economics*, 2017, 132 (1), 435–483.
- East, Chloe N., Annie Hines, Philip Luck, Hani Mansour, and Adrea Velásquez, "The Labor Market Effects of Immigration Enforcement," *Journal of Labor Economics*, 2023.
- **European Commission**, "European Business Cycle Indicators A Closer Look at Labour Shortages across the EU," *Technical Paper 059*, 2022.
- Haltiwanger, John, Ron Jarmin, Robert Kulick, Javier Miranda, Veronika Penciakova, and Cristina Tello-Trillo, "Firm-level Revenue Dataset," CES Technical Notes Series 19-02, Center for Economic Studies, U.S. Census Bureau August 2019.
- Kerr, William R. and William F. Lincoln, "STEM Workers, H-1B Visas, and Productivity in Cities," *Journal of Labor Economics*, 2010, 28 (3), 473–508.
- Kosakow, Jason and Sonya Ravindranath Waddell, "How to Get the Workers," https://www.richmondfed.org/research/regional_economy/regional_matters/2022/ rm_03_31_2022_job_candidate 2022.
- Mahajan, Parag, "Immigration and Business Dynamics: Evidence from U.S. Firms," CESifo Working Paper 9874, CESifo 2022.
- Mitaritonna, Cristina, Gianluca Orefice, and Giovanni Peri, "Immigrants and firms' outcomes: Evidence from France," *European Economic Review*, 2017, *96*, 62 – 82.
- **Orefice, Gianluca and Giovanni Peri**, "Immigration and Worker-Firm Matching," Working Paper 26860, National Bureau of Economic Research March 2020.
- **Orrenius, Pia M. and Madeline Zavodny**, "Help Wanted: Employer Demand for Less-Skilled Temporary Foreign Worker Visas in an Era of Declining Unauthorized Immigration," *RSF: The Russell Sage Foundation Journal of the Social Sciences*, 2020, 6 (3), 45–67.
- **Passel, Jeffrey S. and D'Vera Cohn**, "A Portrait of Unauthorized Immigrants in the United States," Technical Report, Pew Hispanic Center 2009.
- Pekkala Kerr, Sari, William R. Kerr, and William F. Lincoln, "Skilled Immigration and the Employment Structures of US Firms," *Journal of Labor Economics*, 2015, 33 (S1), S147– S186.
- **Peri, Giovanni**, "The Effect of Immigration On Productivity: Evidence from U.S. States," *Review of Economics and Statistics*, 2012, 94 (1), 348–358.
- __, Kevin Shih, and Chad Sparber, "STEM Workers, H-1B Visas, and Productivity in Cities," Journal of Labor Economics, 2015, 33 (S1), S225–S255.
- Pinoti, Paolo, "Clicking on Heaven's Door: The Effect of Immigration Legalization on Crime," American Economic Review, 2017, 107 (1), 138–168.
- Roth, Jonathan and Jiafeng Chen, "Log-like? Identified ATEs defined with zero-valued outcomes are (arbitrarily) scale-dependent," Technical Report 2022.
- Ruggles, Steven, Sarah Flood, Ronald Goeken, Josiah Grover, Erin Meyer, Jose Pacas, and Matthew Sobek, "Integrated Public Use Microdata Series, Current Population Survey: Version 10.0 [dataset]," 2022. Minneapolis, MN: IPUMS, 2022.
- Wasi, N. and A. Flaaen, "Record linkage using Stata: Preprocessing, linking, and reviewing utilities," *Stata Journal*, 2015, 15 (3), 672–697(26).

A Matching Procedure

This procedure is necessitated by a lack of common firm identifiers across the H-2B and Census data. The key variables contained in the H-2B data are: employer name, employer state, and employer city, and employer ZIP code. We include i129 petitions an TLC applications as separate observations so that they can both be linked to a common firm identifier in the Census data. Using pre-processing commands described in Wasi and Flaaen (2015) along with some additional corrections of common mistakes, we clean the names of employers. We then collapse the dataset to the name-state-city-ZIP level.

On the Census side, we link the LBD to the CBPBR using unique, within-year establishment identifiers. The CBPBR also contains name, state, city, and ZIP information for employers. Notably, it includes two name fields and both mailing and physical address for the establishment. Because visa applications are filled out by employers, they may use either the physical or mailing address on their form. We therefore reshape the LBD-CBPBR dataset to have a unique observation for each employer's address. We perform the same pre-processing commands and collapse to the lbdnum-name-state-city-ZIP level. lbdnum is the longitudinal, unique, establishment-level identifier that enables all of the analyses in this paper.

The match proceeds in 6 steps, looping over states (implicitly requiring a match on state), using the reclink2 command (Wasi and Flaaen, 2015):

- 1. Exact matching on all four variables.
- 2. Exact match on ZIP, fuzzy match on employer name1 and city, with more emphasis on name
- 3. Exact match on ZIP, fuzzy match on employer name1 and city, with slightly less emphasis on the name and a higher match score requirement
- 4. Fuzzy match on ZIP, employer name1, and city, with an even higher match score requirement.
- 5. Repeat Steps 2.-4. with employer name2

B Supplemental Figures and Tables

Fiscal Year	DoS Issuances	USCIS Approvals	DoS Denials	Implied Conversion Rate (%)
2007	134,807	179,819		75.0
2008	95,036	107,920		88.1
2009	45,273	79,371		57.0
2010	47,987	70,341		68.2
2011	$51,\!514$	70,339		73.2
2012	$50,\!554$	$64,\!588$		78.3
2013	58,053	70,963	9,981	81.8
2014	68,424	79,258	10,533	86.3
2015	69,984	82,254	9,188	85.1
Total	621,632	804,853		77.2
Mean				78.0

Table B1: Conversion Rate of USCIS Approvals to DoS Issuances

Sources: Department of Homeland Security, Department of State.

Figure B1: Geographic Distribution of January 1, 2018 TLC Applications (per-worker basis)



Notes: Numbers correspond to the ratio of TLC applications to 2018 Q1 employment in a given county. Figure 6 plots raw totals of TLC applications per county.

Source: Department of Labor Office of Foreign Labor Certification Performance Data (numerator), Business Dynamics Statistics (denominator).

Table B2: Top Five Industries and Occupations Listed on Jan 1., 2018 TLC Applications

Par	el A: Industry (NAICS)	
1.	Landscaping Services (561730)	52.6%
2.	Hotels & Motels (721110)	9.4%
3.	"Other" Amusement & Recreation (713990)	3.0%
4.	Fresh & Frozen Seafood Processing (311712)	2.8%
5.	Poured Concrete Foundation & Structure Contractors (238110)	2.5%
Par	el B: Occupation (SOC)	
1.	Landscaping & Groundskeeping (37-3011)	55.5%
2.	Maids & Housekeeping (37-2012)	6.6%
3.	Construction Laborers (47-2061)	3.5%
4.	Amusement & Recreation Attendants (39-3091)	3.4%
5.	Meat, Poultry, and Fish Cutters & Trimmers (51-3022)	2.2%

Source: Department of Labor Office of Foreign Labor Certification Performance Data

	DoL Requests	DoL Requests	DoL Certifications	DoL Certifications
\hat{eta}_{2015}	-0.0167 (0.0763)	-0.0719 (0.0522)	-0.0109 (0.0690)	-0.0624 (0.0413)
\hat{eta}_{2016}	$\begin{array}{c} 0.0121 \\ (0.0865) \end{array}$	$\begin{array}{c} 0.0351 \ (0.0831) \end{array}$	$0.0418 \\ (0.0819)$	$0.0521 \\ (0.0801)$
\hat{eta}_{2018}	-0.1188 (0.1160)	-0.1759 (0.1209)	$0.0004 \\ (0.0531)$	-0.0518 (0.0517)
\hat{eta}_{2019}	0.0781 (0.1084)	0.0844 (0.0850)	0.1262^{*} (0.0740)	$0.1002 \\ (0.0793)$
Controls		\checkmark		\checkmark
<i>p</i> -value: $\beta_{2015} = \beta_{2016} = 0$	0.9108	0.2604	0.7224	0.1562

Table B3: Estimation Results from Equation (1)—DoL Placebo Tests

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. Outcome variables, which label each outcome, are divided by 2017 mean employment. Each specification is estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and who had positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

	All Approvals	All Approvals	Original Tranche	Original Tranche
\hat{eta}_{2015}	-0.0273 (0.0493)	-0.0223 (0.0364)	-0.0044 (0.0472)	-0.0279 (0.0316)
\hat{eta}_{2016}	$0.0509 \\ (0.0340)$	$0.0508 \\ (0.0330)$	-0.0252 (0.0297)	-0.0406 (0.0278)
\hat{eta}_{2018}	$\begin{array}{c} 0.3073^{***} \\ (0.0523) \end{array}$	$\begin{array}{c} 0.2235^{***} \\ (0.0448) \end{array}$	$\begin{array}{c} 0.4341^{***} \\ (0.0497) \end{array}$	$\begin{array}{c} 0.3228^{***} \\ (0.0382) \end{array}$
\hat{eta}_{2019}	$\begin{array}{c} 0.1270^{***} \\ (0.0365) \end{array}$	$\begin{array}{c} 0.1143^{***} \\ (0.0349) \end{array}$	-0.0421 (0.0382)	-0.0591^{*} (0.0355)
Controls		\checkmark		\checkmark
<i>p</i> -value: $\beta_{2015} = \beta_{2016} = 0$	0.0211	0.0524	0.5449	0.3339

Table B4: Estimation Results from Equation (1)—USCIS i129 Approvals ("First Stage")

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. Outcome variables, which title each column, are divided by 2017 mean employment. Each specification is estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and who had positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

	Q1	Q1	Q2	Q2	Q3	Q3	Q4	Q4
\hat{eta}_{2015}	$0.0146 \\ (0.0182)$	$0.0033 \\ (0.0203)$	$\begin{array}{c} 0.0776^{***} \\ (0.0272) \end{array}$	0.0533^{*} (0.0298)	0.0248 (0.0263)	$0.0176 \\ (0.0291)$	$0.0290 \\ (0.0225)$	$0.0305 \\ (0.0250)$
\hat{eta}_{2016}	-0.0033 (0.0150)	-0.0142 (0.0171)	$\begin{array}{c} 0.0713^{***} \\ (0.0208) \end{array}$	$\begin{array}{c} 0.0518^{**} \\ (0.0221) \end{array}$	$\begin{array}{c} 0.0353 \ (0.0233) \end{array}$	$\begin{array}{c} 0.0271 \\ (0.0264) \end{array}$	$\begin{array}{c} 0.0010 \\ (0.0197) \end{array}$	-0.0029 (0.0202)
\hat{eta}_{2018}	-0.0347^{*} (0.0187)	-0.0386^{*} (0.0207)	$\begin{array}{c} 0.2344^{***} \\ (0.0264) \end{array}$	$\begin{array}{c} 0.2077^{***} \\ (0.0264) \end{array}$	$\begin{array}{c} 0.1411^{***} \\ (0.0274) \end{array}$	$\begin{array}{c} 0.1386^{***} \\ (0.0282) \end{array}$	$0.0268 \\ (0.0215)$	$\begin{array}{c} 0.0203 \\ (0.0232) \end{array}$
\hat{eta}_{2019}	-0.0445^{**} (0.0202)	-0.0423^{*} (0.0225)	$\begin{array}{c} 0.0313 \\ (0.0328) \end{array}$	$0.0139 \\ (0.0368)$	$\begin{array}{c} 0.0493 \\ (0.0399) \end{array}$	$0.0386 \\ (0.0450)$	-0.0017 (0.0271)	$\begin{array}{c} 0.0053 \ (0.0304) \end{array}$
Controls		\checkmark		\checkmark		\checkmark		\checkmark
<i>p</i> -value: $\beta_{2015} = \beta_{2016} = 0$	0.5606	0.5567	0.0021	0.0608	0.3163	0.5918	0.2571	0.2494

Table B5: Estimation Results from Equation (1)—Quarterly Employment

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. Outcome variables (employment in the quarter indicated by a given column title) are divided by 2017 mean employment. Each specification is estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and who had positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

	Q1	Q1	Q2	Q2	Q3	Q3	Q4	Q4
\hat{eta}_{2015}	$\begin{array}{c} 0.2297^{**} \\ (0.0924) \end{array}$	$\begin{array}{c} 0.1149 \\ (0.0969) \end{array}$	$\begin{array}{c} 0.5224^{***} \\ (0.1469) \end{array}$	$\begin{array}{c} 0.3357^{**} \\ (0.1644) \end{array}$	$0.2356 \\ (0.1956)$	0.1518 (0.2086)	0.0837 (0.1793)	0.0373 (0.2085)
\hat{eta}_{2016}	$\begin{array}{c} 0.1646^{**} \\ (0.0760) \end{array}$	$\begin{array}{c} 0.1263 \\ (0.0841) \end{array}$	0.2170^{*} (0.1271)	$0.0398 \\ (0.1346)$	0.2748^{*} (0.1513)	0.2785^{*} (0.1682)	$\begin{array}{c} 0.0037 \\ (0.1384) \end{array}$	-0.0471 (0.1509)
\hat{eta}_{2018}	$\begin{array}{c} 0.2031^{**} \\ (0.0859) \end{array}$	$\begin{array}{c} 0.2287^{***} \\ (0.0886) \end{array}$	$\begin{array}{c} 1.560^{***} \\ (0.1570) \end{array}$	$\begin{array}{c} 1.439^{***} \\ (0.1592) \end{array}$	$\begin{array}{c} 1.295^{***} \\ (0.1753) \end{array}$	$\begin{array}{c} 1.332^{***} \\ (0.1747) \end{array}$	$\begin{array}{c} 0.5068^{***} \\ (0.1834) \end{array}$	$\begin{array}{c} 0.5758^{***} \\ (0.1992) \end{array}$
\hat{eta}_{2019}	$\begin{array}{c} 0.0276 \ (0.1358) \end{array}$	$0.1140 \\ (0.1320)$	-0.1608 (0.2093)	-0.2394 (0.2160)	$\begin{array}{c} 0.5871^{**} \\ (0.2516) \end{array}$	$\begin{array}{c} 0.5977^{**} \\ (0.2644) \end{array}$	$0.1963 \\ (0.2219)$	$0.2543 \\ (0.2255)$
Controls		\checkmark		\checkmark		\checkmark		\checkmark
<i>p</i> -value: $\beta_{2015} = \beta_{2016} = 0$	0.0333	0.3042	0.0012	0.0413	0.1875	0.2511	0.8567	0.8583

Table B6: Estimation Results from Equation (1)—Quarterly Payroll

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. Outcome variables (payroll in the quarter indicated by a given column title) are divided by 2017 mean employment. Each specification is estimated on the set of 2,800 single-unit firms that applied to the Department of Labor for a TLC certification on January 1, 2018 and who had positive payroll and employment in Q2 and Q3 of 2017 and 2018. Specifications with controls include state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.

	2HFY log Pay p.w.	2HFY log Pay p.w.	Revenues	Revenues	log Rev. p.w.	log Rev. p.w.	Active	Active
\hat{eta}_{2015}	-0.0129 (0.0118)	-0.0194 (0.0132)	0.6808 (5.255)	-1.546 (2.986)	-0.0348^{*} (0.0198)	-0.0317 (0.0221)	-0.0112 (0.0110)	-0.0055 (0.0114)
\hat{eta}_{2016}	-0.0153 (0.0102)	-0.0181^{*} (0.0109)	-1.600 (3.735)	-2.753 (2.733)	-0.0215 (0.0158)	-0.0275 (0.0176)	-0.0015 (0.0081)	$0.0058 \\ (0.0084)$
\hat{eta}_{2018}	$0.0154 \\ (0.0098)$	$0.0127 \\ (0.0108)$	7.084^{*} (4.140)	9.488^{**} (4.306)	-0.0125 (0.0227)	-0.0221 (0.0205)	$\begin{array}{c} 0.0137^{**} \\ (0.0067) \end{array}$	0.0159^{**} (0.0069)
\hat{eta}_{2019}	-0.0034 (0.0109)	-0.0031 (0.0123)		_	_	_	$\begin{array}{c} 0.0259^{***} \\ (0.0092) \end{array}$	$\begin{array}{c} 0.0306^{***} \\ (0.0098) \end{array}$
Firms Controls	2,800	2,800 √	2,400	2,400 ✓	2,400	2,400 ✓	3,200	3,200 ✓
<i>p</i> -value: $\beta_{2015} = \beta_{2016} = 0$	0.3081	0.2031	0.8648	0.5950	0.2011	0.2418	0.4808	0.3410

Table B7: Estimation Results from Equation (1)—Additional Outcomes

Notes: *** p < 0.01, ** p < 0.05, * p < 0.1. Standard errors, clustered at the firm level, in parentheses. Each column represents a different specification. Outcome variables title each column. The tevenues outcome is divided by 2017 mean employment. Specifications with controls include state-by-year fixed effects, industry-by-year fixed effects and size-quintile-by-year fixed effects.